Author's response to reviews

Title: Bordetella bronchiseptica pneumonia in patient with AIDS.

Authors:

    Michal MG Galeziok (michalgaleziok@gmail.com)
    Ingram IR Roberts (iroberts@stvincents.org)
    Jo-Anne JP Passalacqua (iddocs@hotmail.com)

Version: 4 Date: 26 December 2008

Author's response to reviews:

1. Literature review was added to introduction part.
2. Literature positions was added to reference.